Author's response to reviews

Title: Interleukin-10 polymorphisms in Spanish IgA deficiency patients: a case-control and family study

Authors:

Javier Ortiz (ortizesa@yahoo.es)
Miguel Fernandez-Arquero (mfernandeza hcsc@salud.madrid.org)
Elena Urcelay (eurcelay@hcsc.es)
Raquel Lopez-Mejias (ralome@wanadoo.es)
Antonio Ferreira (aferreira hulp@salud.madrid.org)
Gumersindo Fontan (gfontan hulp@salud.madrid.org)
Emilio G de la Concha (egomez dela hcsc@salud.madrid.org)
Alfonso Martinez (alfmdoncel terra.es)

Version: 3 Date: 19 May 2006

Author's response to reviews:

BMC Medical Genetics
BioMed Central Ltd
Middlesex House, 34-42 Cleveland Street, London W1T 4LB, UK.

May 18th, 2006

Interleukin-10 polymorphisms in Spanish IgA deficiency patients: a case-control and family study

Javier Ortiz, Miguel Fernandez-Arquero, Elena Urcelay, Raquel Lopez-Mejias, Antonio Ferreira, Gumersindo Fontan, Emilio G de la Concha, Alfonso Martinez

We would like to resubmit the final version of this manuscript to be published as a Research Article in BMC Medical Genetics.

Changes made as requested:
Sections headlines are no longer underlined
Materials and Methods has been renamed "Methods"
A "Conclusions" section has been added
"Author's contributions" has been corrected.
Tables are now at the end of the manuscript text

Corresponding author: Alfonso Martinez Doncel, Immunology Department, Hospital Universitario San Carlos, Prof. Martin Lagos s/n, 28040 Madrid, Spain.
Tel 34-91 330 33 47; FAX: 34-91 330 33 44
e-mail: alfmdoncel terra.es